



**Baishideng
Publishing
Group**

7041 Koll Center Parkway, Suite
160, Pleasanton, CA 94566, USA
Telephone: +1-925-399-1568
E-mail: office@baishideng.com
https://www.wjgnet.com

Reviewer 1

Results - Line 120-121 - Remove this sentence as it adds nothing to manuscript.
Discussion - Line 189 - word "instate" should be changed to "determine". - Line 191-195 -
Remove from Discussion and make this statement the Conclusion. Conclusion - Remove
the words in the Conclusion and make Line 191-195 the new Conclusion,. Reference 10
has no year of publishing.

Dear reviewer, please find below the changes made as suggested

Results - Line 120-121 - Remove this sentence as it adds nothing to manuscript. -
REMOVED
Discussion - Line 189 - word "instate" should be changed to "determine". - CHANGED
Line 191-195 - Remove from Discussion and make this statement the Conclusion.
Conclusion - Remove the words in the Conclusion and make Line 191-195 the new
Conclusion,. DONE AS SUGGESTED
Reference 10 has no year of publishing. - ADDED



**Baishideng
Publishing
Group**

7041 Koll Center Parkway, Suite
160, Pleasanton, CA 94566, USA
Telephone: +1-925-399-1568
E-mail: office@baishideng.com
<https://www.wjgnet.com>

Reviewer 2

This is a nice topic to write about. unfortunately the language quality is poor so it makes reading and enjoying it a bit difficult Since metal implants are hardly in use nowadays , the relevance of this paper is in doubt Despite all I would recommend to consider its publishing after a major revision

Dear reviewer, we are thankful to you for your inputs. The language quality has been improved and enhanced in the latest version of the manuscript. Please find the enclosed version with highlighted changes.